Author’s response to reviews

Title: Short-term effects of plain cigarette packaging on smoking behaviour and attitudes: A randomised controlled study

Authors:

Olivia M Maynard (olivia.maynard@bristol.ac.uk)
Ute Leonards (ute.leonards@bristol.ac.uk)
Angela S Attwood (angela.attwood@bristol.ac.uk)
Linda Bauld (linda.bauld@stir.ac.uk)
Lee Hogarth (l.hogarth@exeter.ac.uk)
Marcus R Munafò (marcus.munafo@bristol.ac.uk)

Version: 2 Date: 5 November 2014

Author’s response to reviews:

Dear Professor Pafitis,

Thank you for quickly assessing our manuscript for compliance with the journal style. Please find details of the changes we have made to our manuscript below. We have made each of the changes recommended by the Journal Editorial Office.

1. We have included email addresses for all authors on the title page of the manuscript.

2. We have added page numbers and line numbers.

3. We have added information about the trial registry to the title page of the manuscript (Trial registration: Current Controlled Trials ISRCTN52982308. Registered 27 June 2013).

4. We have added further information about the specific contributions of the authors in the Author’s Contributions section.

5. We have added the two tables to the end of the manuscript after the results section and added the table headings.

Please let us know if any further changes are required. We look forward to hearing from you again with peer review comments on our manuscript,

Yours sincerely,

Olivia M. Maynard